Author’s response to reviews

Title: Transient symptomatic hyperglycaemia secondary to inhaled fluticasone propionate in a young child: a case report

Authors:

Susanna Esposito (Susanna.Esposito@unimi.it)
Mara Lelii (mara.lelii87@gmail.com)
Nicola Principi (nicola.principi@unimi.it)

Version: 2 Date: 27 Dec 2015

Author’s response to reviews:

Dear Editor,

Thank you for considering our manuscript PULM-D-15-00125R1 entitled "Transient symptomatic hyperglycaemia secondary to inhaled fluticasone propionate in a young child", which we submit for publication as Case Report in BMC Pulmonary Medicine.

Please find attached the revised manuscript and our replies to Editor’s comments and recommendations. The authors, all of whom contributed significantly to the manuscript and declare that they have no potential conflict of interest, have seen and approved the final version of the manuscript.

We declare that the text has been reviewed by a native English speaker with appropriate knowledge of the subject matter. We hope that you will now find the paper suitable for publication in BMC Pulmonary Medicine.

Yours faithfully,

Susanna Esposito, MD
Pediatric Highly Intensive Care Unit
Editor

Thank you very much for the appreciation of our manuscript. We have modified it as suggested.

1. The term “hyperglycaemia” is now spelled correctly in the Title (p. 1).

2. The absence of evidence of causation of this interesting and clinically relevant risk has been highlighted in the Abstract (p. 3) and the Conclusions (p. 7).

3. Our paper is already in line with all Editorial requests.